Author’s response to reviews

Title: Small-cell Neuroendocrine Carcinoma of the Ileum: Case Report

Authors:
Jong Eun Lee (elduke00@schmc.ac.kr)
Sung Hoon Hong (soondaeman00@naver.com)
Hae Il Jung (gs2834@schmc.ac.kr)
Myoung Won Son (mwson@schmc.ac.kr)
Tae sung Ahn (eyetoeye@schmc.ac.kr)
Sun Wook Han (cheia@schmc.ac.kr)
JunHun Cho (jununiu@schmc.ac.kr)

Version: 4 Date: 16 Aug 2019

Author’s response to reviews:

1. It was changed other phrase. It minimized overlap section.
2. It was specified consent to publish.
3. It was changed introduction to background
4. I have been completed declaration sections.

Thank you